Author's response to reviews

Title: Dexmedetomidine is neuroprotective in an in vitro model of traumatic brain injury

Authors:

Marc Schoeler (marc.schoeler@wth-aachen.de)
Philip D Loetscher (ploetscher@ukaachen.de)
Rolf Rossaint (rrossaint@ukaachen.de)
Astrid V Fahlenkamp (afahlenkamp@ukaachen.de)
Georg Eberhardt (gfeberhardt@ukaachen.de)
Steffen Rex (srex@ukaachen.de)
Joachim Weis (jweis@ukaachen.de)
Mark Coburn (mcoburn@ukaachen.de)

Version: 3 Date: 1 November 2011

Author's response to reviews:

Dear Editorial Board,

On behalf of my co-authors, I am submitting the enclosed material for possible publication in BMC Neurology. We want to share these interesting in vitro data of a widely used drug in intensive care in the states and which has just become approved in Europe some weeks ago. The work has not been submitted for publication nor has it been published in whole or in part elsewhere. I attest to the fact that all authors listed on the title page have read the manuscript, have confirmed the validity and legitimacy of the data and its interpretation, and agree to its submission to BMC Neurology. I confirm that all authors have made a substantial contribution to the conduct of the study or preparation of the manuscript.

I am looking forward to hear from you!

Thank you very much!

Yours sincerely

Mark Coburn